Author’s response to reviews

Title: Vitamin B6 deficiency and diseases in elderly - a study in nursing homes

Authors:

Ida K Kjeldby (ida.k.kjeldby@gmail.com)
Gunvor S Fosnes (gunvor.fosnes@gmail.com)
Solveig C Ligaarden (solveig.ligaarden@sykehuset-innlandet.no)
Per G Farup (per.farup@ntnu.no)

Version: 7 Date: 17 December 2012

Author’s response to reviews: see over
The Editor,
BMC Geriatrics

Dear Editor,

I refer to your mail of December 17th, 2012. Changes in our manuscript “Vitamin B6 deficiency and diseases in elderly – a study in nursing homes” have been performed according to your recommendations.

Your information about the origin of the word “predict” is for sure correct, but there are numerous examples of words not used according to the original meaning, also in medicine and science. The discussion we have had is detrimental for the reputation of BMC Geriatrics and BMC in general.

Sincerely yours
On behalf of the authors

Per G Farup MD, Ph.D
Professor, Faculty of Medicine, Norwegian University of Science and Technology.
Associate Editor, BMC Gastroenterology.